\begin{table}%T2
	\caption{\label{tab.jrnl}Journal of interferometric data. 
	$K$~band (FLUOR) data for $\alpha$~Orionis have been presented in \cite{cit.perrin2002aori}, for SW~Virginis in \cite{cit.perrin2003a}, and for R~Leonis in \cite{cit.perrin1999}. 
	All $L$~band (TISIS) data were taken from \cite{cit.chagnon2002}. 
	The ISI data points for $\alpha$~Orionis and R~Leonis were extracted from \cite{cit.danchi1994} using the DEXTER online tool at \cite{ads}. No exact position angle was given along with the ISI data, therefore we indicated the basic geographic orientation of the baselines used. 
	Where known for the objects, the phase $\Phi$ of the visual variability was indicated. 
	}
%\centering
\par
	\begin{tabular}{cccc|cccc}
	\hline\hline
  Object  &  Spatial Frq.  &  Visibility  &  Pos. angle  &  Object  &  Spatial Frq.  &  Visibility  &  Pos. angle  \\
    &  ($10^{5}~{\rm rad}^{-1}$)  &    &   (deg)   &    &  ($10^{5}~{\rm rad}^{-1}$)  &    &   (deg)   \\
	\hline
$\alpha$~Ori  &  \multicolumn{3}{l|}{FLUOR}  &          R~Leo  &  \multicolumn{3}{l}{FLUOR, Apr. 1996, $\Phi = 0.24$}  \\
  &  46.41  &  0.2232  $\pm$  0.0018  &  85.91  &	   &  65.76  &  0.2723  $\pm$  0.0040  &  86.9  \\ 
  &  46.82  &  0.2045  $\pm$  0.0020  &  80.47  &	   &  65.76  &  0.2718  $\pm$  0.0043  &  92.5  \\
  &  48.14  &  0.1847  $\pm$  0.0024  &  71.96  &	   &  66.38  &  0.2723  $\pm$  0.0041  &  80.3  \\
  &  50.39  &  0.1253  $\pm$  0.0016  &  63.01  &	   &  69.82  &  0.2373  $\pm$  0.0033  &  65.4  \\
  &  88.30  &  0.0949  $\pm$  0.0047  &  69.89  &	   &  71.57  &  0.2127  $\pm$  0.0035  &  60.7  \\
  &  90.92  &  0.0843  $\pm$  0.0024  &   --    &	   &  \multicolumn{3}{l}{FLUOR, Mar. 1997, $\Phi = 0.28$}  \\
  & 133.99  &  0.0510  $\pm$  0.0010  &  79.06  &	   &  47.94  &  0.4556  $\pm$  0.0041  &  86.0  \\
  &  \multicolumn{3}{l|}{TISIS, March 2000}  &              &  50.08  &  0.4248  $\pm$  0.0036  &  68.5  \\
  &  35.99  &  0.517  $\pm$  0.015  &   88.85  &           &  66.89  &  0.1934  $\pm$  0.0036  &  77.0  \\
  &  35.99  &  0.528  $\pm$  0.014  &   89.49  &           &  68.27  &  0.1751  $\pm$  0.0029  &  70.9  \\
  &  36.34  &  0.521  $\pm$  0.011  &   81.08  &           &  70.69  &  0.1656  $\pm$  0.0050  &  63.2  \\
  &  53.32  &  0.085  $\pm$  0.001  &  108.69  &           &  70.96  &  0.1761  $\pm$  0.0071  &  62.5  \\
  &  53.71  &  0.081  $\pm$  0.001  &  110.40  &           &  91.07  &  0.1255  $\pm$  0.0096  &  71.2  \\
  &  \multicolumn{3}{l|}{ISI, Oct. 1988--Oct. 1989}  &   &  97.21  &  0.1188  $\pm$  0.0060  &  58.5  \\
  &  1.64  &  0.54  $\pm$  0.04  &  90  &                  & 150.62  &  0.0770  $\pm$  0.0032  &  63.1  \\
  &  1.82  &  0.57  $\pm$  0.02  &  90  &                  &  \multicolumn{3}{l}{TISIS, Mar. 2000, $\Phi = 0.81$}  \\
  &  2.24  &  0.61  $\pm$  0.01  &  90  &                  &  39.29  &  0.515  $\pm$  0.011  &  168.48  \\
  &  2.36  &  0.65  $\pm$  0.01  &  90  &                  &  39.46  &  0.528  $\pm$  0.011  &   67.44  \\
  &  2.51  &  0.62  $\pm$  0.02  &  90  &                  &  40.35  &  0.504  $\pm$  0.013  &   62.78  \\
  &  2.75  &  0.59  $\pm$  0.01  &  90  &                  &  52.76  &  0.249  $\pm$  0.002  &   86.66  \\
  &  2.96  &  0.59  $\pm$  0.01  &  90  &                  &  52.91  &  0.253  $\pm$  0.003  &   96.07  \\
  &  3.26  &  0.56  $\pm$  0.02  &  90  &                  &  53.69  &  0.258  $\pm$  0.003  &  103.86  \\
  &  3.53  &  0.60  $\pm$  0.01  &  90  &                  &  53.88  &  0.245  $\pm$  0.002  &  105.20  \\
  &  \multicolumn{3}{l|}{ISI, Oct.--Nov. 1992}  &        &  53.94  &  0.251  $\pm$  0.003  &  105.55  \\
  &  6.91  &  0.58  $\pm$  0.01  &  123  &                 &  54.04  &  0.246  $\pm$  0.002  &  106.15  \\
  &  7.39  &  0.59  $\pm$  0.01  &  123  &                 &  54.10  &  0.248  $\pm$  0.002  &  106.58  \\
  &  7.57  &  0.56  $\pm$  0.02  &  123  &		   &  54.12  &  0.227  $\pm$  0.002  &  106.74  \\
  &  7.96  &  0.54  $\pm$  0.02  &  123  &		   &  54.85  &  0.229  $\pm$  0.003  &  110.85  \\
  &  8.67  &  0.56  $\pm$  0.02  &  123  &		   &  \multicolumn{3}{l}{ISI, Oct. 1988, $\Phi \approx 0.45$}  \\
  &  \multicolumn{3}{l|}{ISI, Nov. 1989--Oct. 1992}  &   &  2.44  &  0.86  $\pm$  0.05  &  90  \\
  &   8.97  &  0.61  $\pm$  0.02  &  113?  &		   &  2.54  &  0.83  $\pm$  0.04  &  90  \\
  &   9.18  &  0.58  $\pm$  0.01  &  113  &		   &  \multicolumn{3}{l}{ISI, Oct. 1990, $\Phi  = 0.81$}  \\
  &   9.30  &  0.60  $\pm$  0.02  &  113  &		   &  7.30  &  0.55  $\pm$  0.06  &  113  \\
  &   9.78  &  0.56  $\pm$  0.03  &  113  &		   &  7.51  &  0.63  $\pm$  0.03  &  113  \\
  &   9.96  &  0.55  $\pm$  0.02  &  113  &		   &  8.02  &  0.53  $\pm$  0.04  &  113  \\
  &  10.22  &  0.54  $\pm$  0.02  &  113  &		   &  8.23  &  0.65  $\pm$  0.05  &  113  \\
  &  10.55  &  0.56  $\pm$  0.02  &  113  &		   &  8.62  &  0.59  $\pm$  0.03  &  113  \\
  &  10.70  &  0.59  $\pm$  0.02  &  113  &		   &  9.10  &  0.54  $\pm$  0.03  &  113  \\
  &  10.91  &  0.54  $\pm$  0.01  &  113  &		   &  9.30  &  0.65  $\pm$  0.06  &  113  \\
  &  10.97  &  0.58  $\pm$  0.02  &  113  &		   &  9.61  &  0.53  $\pm$  0.03  &  113  \\
  &  11.30  &  0.54  $\pm$  0.01  &  113  &		   &  9.81  &  0.59  $\pm$  0.03  &  113  \\
  &  11.45  &  0.48  $\pm$  0.02  &  113  &		   & 10.02  &  0.60  $\pm$  0.02  &  113  \\
  &  11.57  &  0.55  $\pm$  0.02  &  113  &		   & 10.20  &  0.56  $\pm$  0.03  &  113  \\
  &  11.69  &  0.50  $\pm$  0.02  &  113  &		   & 10.32  &  0.57 \mbox{\rule[0ex]{6ex}{0ex}} &  113  \\
  &  11.81  &  0.52  $\pm$  0.01  &  113  \\
\cline{1-4}
SW~Vir  &  \multicolumn{3}{l|}{FLUOR, May 2000, $\Phi = 0.75$}  \\
  &  64.29  &  0.697  $\pm$  0.010  &  61.31  \\
  &  81.89  &  0.522  $\pm$  0.014  &  93.93  \\
  &  81.93  &  0.527  $\pm$  0.014  &  94.23  \\
  &  \multicolumn{3}{l|}{TISIS, Mar. 2000, $\Phi = 0.35$}  \\
  &  32.63  &  0.830  $\pm$  0.012  &  76.01  \\
  &  32.73  &  0.848  $\pm$  0.012  &  75.39  \\
  &  46.80  &  0.706  $\pm$  0.008  &  98.57  \\
\hline
\end{tabular}
\end{table}